Author’s response to reviews

Title: A study protocol for an mHealth, multi-centre randomized control trial to promote use of postpartum contraception amongst rural women in Punjab, Pakistan

Authors:
Xaher Gul (xaher.gul@mariestopes.org.pk)
Waqas Hameed (waqas.hameed1@gmail.com)
Sharmeen Hussain (sharmeenhussain90@gmail.com)
Ishaque Sheikh (ishaque.sheikh@mariestopes.org.pk)
Junaid-ur-Rehman Siddiqui (junaidrehman1994@hotmail.com)

Version: 3 Date: 17 Jul 2019

Author’s response to reviews:

1. Please indicate the corresponding author on the title page of your manuscript.

Response: Junaid-ur-Rehman Siddiqui has been highlighted as the corresponding author

2. Please remove the appendix as the questionnaire is included as an additional file and the consent form is not needed for publication.

Response: Removed as suggested

3. Please specify what age parental consent would be required, in the Ethics approval and consent to participate section (for example, parental consent was obtained for participants under the age of 16)

Response: Specified under the concerned section on Page 19

4. We note that IH is included on the title page, but they are missing from the author's contributions section. Please ensure that all authors are included in both sections. If there has been a change in authorship, we use a standardised form which we would request you and your co-authors to complete. The change in authorship form can be found from the link on the following page: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5
Response: IS on the title page was misspelled in the authors' contribution as MI. This has been corrected, and there are no changes to the authorship of the manuscript.

5. Please acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials.

Response: There are no acknowledgements to be declared from the authors.

6. Please proofread and ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes, colored text, or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Response: Done as suggested.